

GOPEN ACCESS

Citation: Shearer FM, Huang Z, Weiss DJ, Wiebe A, Gibson HS, Battle KE, et al. (2016) Estimating Geographical Variation in the Risk of Zoonotic *Plasmodium knowlesi* Infection in Countries Eliminating Malaria. PLoS Negl Trop Dis 10(8): e0004915. doi:10.1371/journal.pntd.0004915

Editor: Thomas S. Churcher, Imperial College London, UNITED KINGDOM

Received: May 12, 2016

Accepted: July 20, 2016

Published: August 5, 2016

Copyright: © 2016 Shearer et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work was principally funded by a Wellcome Trust grant (http://www.wellcome.ac.uk; 095066, SIH, KEB, NG, HSG, CLM) and a University of Oxford Rhodes Scholarship (http://www. rhodeshouse.ox.ac.uk; FMS). Additional salary support was provided by the Bill & Melinda Gates Foundation through the Vector Borne Disease Network (https:// www.vecnet.org; ZH, AW, CLM) and directly (http:// www.gatesfoundation.org; OPP1068048, DJW, HSG, SB, PWG; OPP1106023, DJW, HSG, PWG; RESEARCH ARTICLE

Estimating Geographical Variation in the Risk of Zoonotic *Plasmodium knowlesi* Infection in Countries Eliminating Malaria

Freya M. Shearer¹, Zhi Huang², Daniel J. Weiss², Antoinette Wiebe¹, Harry S. Gibson², Katherine E. Battle², David M. Pigott³, Oliver J. Brady¹, Chaturong Putaporntip⁴, Somchai Jongwutiwes⁴, Yee Ling Lau⁵, Magnus Manske^{6,7}, Roberto Amato^{6,7,8}, Iqbal R. F. Elyazar⁹, Indra Vythilingam⁵, Samir Bhatt^{2,10}, Peter W. Gething², Balbir Singh¹¹, Nick Golding^{1,12}, Simon I. Hay^{1,3}, Catherine L. Moyes^{1*}

 Spatial Ecology & Epidemiology Group, Oxford Big Data Institute, Li Ka Shing Centre for Health Information and Discovery, University of Oxford, Oxford, United Kingdom, 2 Malaria Atlas Project, Oxford Big Data Institute, Li Ka Shing Centre for Health Information and Discovery, University of Oxford, Oxford, United Kingdom, 3 Institute of Health Metrics and Evaluation, University of Washington, Seattle, Washington, United States of America, 4 Molecular Biology of Malaria and Opportunistic Parasites Research Unit, Department of Parasitology, Faculty of Medicine, Chulalongkorn University, Bangkok, Thailand, 5 Department of Parasitology, Faculty of Medicine, University of Malaya, Kuala Lumpur, Malaysia, 6 Wellcome Trust Sanger Institute, Hinxton, United Kingdom, 7 Medical Research Council (MRC) Centre for Genomics and Global Health, University of Oxford, Oxford, United Kingdom, 8 Wellcome Trust Centre for Human Genetics, University of Oxford, Oxford, United Kingdom, 9 Eijkman-Oxford Clinical Research Unit, Jakarta, Indonesia, 10 Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom, 11 Malaria Research Centre, Universiti Malaysia Sarawak, Kuching, Sarawak, Malaysia, 12 Department of BioSciences, University of Melbourne, Parkville, Victoria, Australia

* catherinemoyes@gmail.com

Abstract

Background

Infection by the simian malaria parasite, *Plasmodium knowlesi*, can lead to severe and fatal disease in humans, and is the most common cause of malaria in parts of Malaysia. Despite being a serious public health concern, the geographical distribution of *P. knowlesi* malaria risk is poorly understood because the parasite is often misidentified as one of the human malarias. Human cases have been confirmed in at least nine Southeast Asian countries, many of which are making progress towards eliminating the human malarias. Understanding the geographical distribution of *P. knowlesi* is important for identifying areas where malaria transmission will continue after the human malarias have been eliminated.

Methodology/Principal Findings

A total of 439 records of *P. knowlesi* infections in humans, macaque reservoir and vector species were collated. To predict spatial variation in disease risk, a model was fitted using records from countries where the infection data coverage is high. Predictions were then made throughout Southeast Asia, including regions where infection data are sparse. The resulting map predicts areas of high risk for *P. knowlesi* infection in a number of countries that are forecast to be malaria-free by 2025 (Malaysia, Cambodia, Thailand and Vietnam)

OPP1093011, DMP; OPP1119467, OJB), the Wellcome Trust (http://www.wellcome.ac.uk/; 098051, MM, RA; B9RZGS0, IRFE), a McKenzie fellowship from the University of Melbourne (http://research. unimelb.edu.au/; NG), the University Malaysia Sarawak (http://www.unimas.my/; 01(TD03)/1003/2013 (01), BS), the Thailand Research Fund (http://www.trf. or.th/eng/; RSA5480008; CP), the Thailand Government Research Budget (http://www.thaigov.go. th/; GRB-APS 12593011; SJ), a University of Malaya High Impact Research (HIR) Grant UM-MOHE from the Ministry of Higher Education, Malaysia (http://hir. um.edu.my/; UM.C/HIR/MOHE/MED/16, YLL), and an MRC/DFID Career Development Fellowship (https:// www.mrc.ac.uk/; K00669X, PWG). The funders had no role in the study design, data collection or analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

as well as countries projected to be eliminating malaria (Myanmar, Laos, Indonesia and the Philippines).

Conclusions/Significance

We have produced the first map of *P. knowlesi* malaria risk, at a fine-scale resolution, to identify priority areas for surveillance based on regions with sparse data and high estimated risk. Our map provides an initial evidence base to better understand the spatial distribution of this disease and its potential wider contribution to malaria incidence. Considering malaria elimination goals, areas for prioritised surveillance are identified.

Author Summary

Plasmodium knowlesi is a malaria parasite found in wild monkey populations and transmitted from this animal reservoir to humans *via* infected mosquitoes. It causes severe and fatal disease in humans, and is the most common cause of malaria in parts of Malaysia. The geographical distribution of this disease is largely unknown because it is often misdiagnosed as one of the human malarias. Human malaria parasites are primarily transmitted between humans *via* mosquitoes and are not frequently transmitted from other animals to humans. Many countries in Southeast Asia, where *P. knowlesi* infections have been reported, are making progress towards eliminating the human malarias. Understanding the geographical distribution of *P. knowlesi* is important for identifying areas where malaria transmission will continue after the human malarias have been eliminated. In locations that have high volumes of *P. knowlesi* infection data, we modelled patterns of variation in the data linked to environmental predictors, and used this to estimate *P. knowlesi* infection risk in locations where data is lacking. The resulting map represents an initial evidence-base for identifying areas of human disease risk that should be prioritized for surveillance, particularly in the context of malaria elimination in the region.

Introduction

Malaria cases caused by the simian parasite, *Plasmodium knowlesi*, have been identified in at least nine Southeast Asian countries. In many parts of Malaysia, this parasite is the most common cause of malaria [1] and can lead to severe and fatal disease [2–4]. Despite the potential severity of infection [1, 4–8], diagnostics that identify *P. knowlesi* are not routinely used. Unless blood samples are tested using expensive nested PCR-based diagnostics, cases of *P. knowlesi* are often misdiagnosed by microscopy as one of the human malarias, principally *P. malariae* or *P. falciparum* [9–11].

Plasmodium knowlesi infection is routinely considered as a potential causal pathogen of malaria cases in three countries: Malaysia, Brunei and Singapore (hereafter a region referred to as MBS), the latter two having already eliminated the human malarias. *Plasmodium knowlesi* malaria cases have also been reported in Cambodia, Indonesia, Myanmar, the Philippines, Thailand and Vietnam [8, 12, 13], but sampling has been limited and the full geographical extent of disease risk across most of the region, including within these countries, is unknown.

Understanding the geographical distribution of *P. knowlesi* is important to identify areas where residual malaria transmission could remain once the human malarias, namely *P. falciparum*, *P. vivax*, *P. malariae* and *P. ovale*, have been eliminated [14]. Human malaria parasites are primarily transmitted between humans *via* mosquitoes and are not frequently transmitted

from other animals to humans. Many countries in Southeast Asia, including Malaysia, the Philippines, Thailand and Vietnam, are currently in the process of eliminating the human malarias [15]. Current control measures that reduce these malarias include mass anti-malarial drug administration, the provision of insecticide-impregnated bed nets (ITNs) and indoor residual spraying of houses (IRS). These control measures do not, however, target transmission of the parasite within populations of the reservoir host species so *P. knowlesi* populations will not be eliminated. Further, ITNs and IRS are unlikely to offer the same degree of personal protection to humans, or community protection through reductions in mosquito longevity, since the vectors for *P. knowlesi* bite and rest outdoors [16]. If the presence of *P. knowlesi* is not considered when elimination strategies are developed, the impact of elimination measures and reduction in overall malaria cases in these areas will not match projections.

In this study, we produced the first map of the geographical distribution of *P. knowlesi* malaria, using a niche modelling approach previously applied to the mapping of other vectorborne and zoonotic diseases, including dengue [17], the Leishmaniases [18], Ebola virus disease [19], Lassa fever [20], Marburg virus disease [21], Crimean-Congo hemorrhagic fever [22], and Zika virus [23]. Niche models are able to combine information on locations where diseases have been recorded with geographic data on environmental and socioeconomic factors hypothesized to affect disease transmission [24]. Once the model has been fitted, the potential presence of the disease can be predicted in regions where it has yet to be reported.

To identify regions at risk from a disease with reservoirs in multiple host species, transmitted by multiple vector species, this modelling approach needs to be further refined. The spatial distribution of such diseases is restricted to locations where all species required for transmission coincide, so it is important to consider the distributions of these species [25]. The work presented here builds on previous work that assessed the evidence for the limits of transmission [12]. Here we refine those spatial limits and investigate the variation in risk within them. We recently defined the fine-scale species distributions of the known and putative reservoirs and vectors of *P. knowlesi* [26], including the main macaque species identified as natural hosts of *P.* knowlesi, Macaca fascicularis and M. nemestrina [27–31], and several anopheline mosquito species, all from the Leucosphyrus Group, implicated as vectors of *P. knowlesi* [16, 32–35]. Our maps of these species are useful for defining the limits of zoonotic transmission, but an index of disease risk cannot be extrapolated directly from reservoir/vector maps. While co-occurrence of reservoir and vector species involved in a zoonotic disease system is necessary for transmission, it is not always sufficient, as many other factors contribute. We have developed a model that incorporates the relationships between P. knowlesi infection and the distributions of the reservoir and vector species, along with a range of other potential risk factors, to produce fine-scale evidence-based predictions of relative zoonotic *P. knowlesi* transmission risk.

The final output provides an initial map that aims to identify locations where disease surveillance and epidemiological investigations would be most informative to improve our understanding of disease risk.

Methods

Overview

A schematic of the process we followed is given in Fig 1. We collated and geo-positioned records of *P. knowlesi* infections in humans, and reservoir and vector species, from a variety of sources. The study area was a rectangle encompassing the locations of confirmed or putative *P. knowlesi* infections plus a minimum buffer zone of 300 km, giving an area from northeast Bangladesh to southwest Papua New Guinea. A map predicting the human risk of *P. knowlesi* malaria at every square (pixel) in a 5 km × 5 km grid was generated using an ensemble of





doi:10.1371/journal.pntd.0004915.g001

boosted regression tree (BRT) models to carry out a niche modelling analysis. The model used the database of geo-positioned occurrence points for *P. knowlesi* infections combined with 19 gridded datasets of environmental and socio-economic explanatory covariates as well as probabilistic species distributions for *M. nemestrina*, *M. fascicularis* and the Leucosphyrus Group.

Datasets comprising ad-hoc reports of disease occurrence (as opposed to data from planned region-wide surveys) are subject to spatial bias in reporting rates, which if unaccounted for may result in elevated risk predictions in the areas most likely to report [<u>36</u>]. Reporting bias is likely to be more pronounced for *P. knowlesi* malaria since significant resources are required to accurately diagnose infection and *P. knowlesi* infection is not routinely considered a possible cause of malaria in the region, with the exception of MBS.

A model was therefore fitted using data only from MBS (the model training region), where we could account for reporting bias through our selection of background data. This model was then used to predict the human risk of *P. knowlesi* infection across Southeast Asia. To assess the model's predictive capacity outside its training region, we tested its performance on a set of disease presence and absence records from locations outside MBS. We also generated a

multivariate environmental similarity surface to identify regions where the model was required to extrapolate to environments not found within MBS, and therefore evaluate the appropriateness of inferring risk in those regions.

Occurrence dataset

Records of *P. knowlesi* presence or absence for Southeast Asia were obtained from reports in the published literature from 2004 to 2015, which were validated through personal communications with the authors to confirm details. Each presence record contains the coordinates of a point location or an area greater than 25 km² (polygons) where a human, macaque or mosquito infection was confirmed using specific diagnostics that are able to distinguish *P. knowlesi* from the other *Plasmodium spp.* Presence points were excluded if *P. knowlesi* presence in the surrounding area (within 300 km radius) was not verified by a second, independent laboratory. Each absence record contains the coordinates for a site where an appropriate diagnostic for *P. knowlesi* was used, but no infections were found in a sample size of at least 500 individuals, or where no malaria cases were reported across an administrative division in 2012 [<u>37</u>]. Further details regarding data assemblage are included in the Supporting Information along with the complete data (<u>S1</u> and <u>S2</u> Files).

Explanatory covariates

Nineteen 5 km × 5 km gridded data surfaces of a range of environmental and socio-economic factors, along with predicted reservoir and vector species distributions, were used as explanatory covariates (Table 1). No prior assumptions were made about the nature of any relationships between these covariates and disease risk. Further details regarding the construction of each covariate data surface is provided in the Supporting Information along with plots of each surface (S1 File, and S1 and S2 Figs).

Model fitting

To carry out the niche mapping analysis, we fitted an ensemble of boosted regression tree (BRT) models using the gbm R package [47]. This BRT approach has the ability to fit complex

Covariate	Data source
Open shrublands, woody savannas, savannas, grasslands, wetlands, croplands, and cropland mosaics land cover classes (proportional)	MODIS land cover product [38]
Intact forest cover (proportional)	MODIS land cover product [38] Intact Forest Landscape [39]
Disturbed forest cover (proportional)	MODIS land cover product [38] Intact Forest Landscape [39]
Elevation	Shuttle Radar Topography Mission [40]
Temperature suitability index for <i>Plasmodium falciparum</i> transmission	Gething et al. 2011 [41]
Tasseled cap wetness, a measure of surface moisture (mean and standard deviation)	Gap-filled MODIS satellite data [42, 43]
Tasseled cap brightness, a measure of moisture on bare surfaces (standard deviation)	Gap-filled MODIS satellite data [42, 43]
Human population density	WorldPop [44] and Gridded Population of the World [45]
Urban accessibility	European Commission Joint Research Centre [46]
Species distributions for <i>Macaca fascicularis</i> , <i>M. nemestrina</i> and the Leucosphyrus Group	Moyes et al. 2016 [26]

Table 1. Explanatory covariates included in the analysis and their data source.

doi:10.1371/journal.pntd.0004915.t001

nonlinear responses including high-dimensional interactions between explanatory variables [48], has been shown to have high predictive accuracy [24] and has been previously applied to disease distribution mapping [17-23].

Boosted regression trees combine two algorithms: regression trees (which repeatedly split the data into two groups using a randomly selected predictor variable for each split) and boosting (which additively fits trees to the data, gradually prioritizing poorly modelled data to produce a set of trees that maximally reduce the loss function), to examine and quantify the relationship between explanatory variables and the response data [48]. The core setup used has been described previously [17, 19, 48]. The changes made to the method for the work presented here addressed sampling bias in the infection reports within MBS, incorporated host and vector data, allowed temporal changes in land cover to be incorporated, and improved handling of polygon data.

Rather than exclusively using synoptic (averaged across time) covariate values for each of the occurrence locations irrespective of the occurrence date, we incorporated annual data surfaces describing land cover and reservoir and vector distributions from 2001 to 2012. This was necessary to account for the substantial changes in land cover that have occurred in the region over the study period due to deforestation [49], which is hypothesized to have impacted the distribution of the reservoir and vector species of *P. knowlesi* [50]. Using the annual land cover data surfaces, disease occurrence data collected between 2001 and 2012 were matched with covariate values for the relevant year; most data points (76%) fell within this time period. Covariate values for 2001 were used for occurrence data prior to this date and covariate values for 2012 were used for post-2012 data. Final predictions were made to the most contemporary covariate values available to represent the current distribution of disease risk.

We used a binomial likelihood function for the BRT model in order to incorporate both presence and absence records. Whilst records of disease absence are highly informative, they are rare because they require significantly greater sampling effort to ensure their reliability compared to presence data [51], especially for diseases like *P. knowlesi* malaria where the appropriate diagnostics are rarely used. We therefore supplemented the dataset with a large number of background records to represent areas where the disease has not been reported within MBS. Six thousand background points were generated in total [51] with the same proportion of human, macaque and mosquito background points as the presence dataset. It has been demonstrated that predictive accuracy of presence-background niche models can be improved by selecting background data with similar spatial bias to the occurrence records [36]. Human infection background points were generated by randomly sampling across MBS, biased towards human population density, since more populous areas have a greater probability of reporting at least one case. This method was also used to generate background points for the mosquito infection data since all studies that looked for P. knowlesi infections in vector species, selected study locations based on the presence of human P. knowlesi cases in the immediate vicinity. Background points for the macaque infection data were randomly sampled from a macaque occurrence and mammal survey dataset that reflected the bias in locations chosen for macaque studies [26]. Covariate values for the specific times and locations of the background data were then extracted. Prior to covariate extraction, human and vector background points were assigned a year randomly sampled from the temporal distribution of presence points for each species.

Since the occurrence dataset included data from humans, macaques and mosquitoes, a joint model was fitted for human, macaque and mosquito hosts that enabled all available infection data to be leveraged, whilst not constraining the model to assume that the distribution of infection risk would be identical for all three host organisms. As BRTs can fit high-dimensional interactions, the joint model is able to fit different environmental responses for each host organism, or if there is no difference in the signal, to fit the same response for all of them.

To increase the robustness of model predictions and quantify model uncertainty, we fitted an ensemble of 500 BRT models (sub-models), each trained to a separate bootstrap dataset randomly sampled with replacement from the complete presence/background dataset. To incorporate uncertainty in the precise location of infection for polygon occurrence records, each bootstrap randomly selected a 5 km × 5 km pixel within each polygon. Further information on model fitting can be found in the Supporting Information (S1 File) and the R code used to carry out the analysis is freely available at (https://github.com/fshearer/pk_parasite).

Model prediction and evaluation

To generate the final prediction map, a mean predicted value of suitability for infection was calculated across the 500 sub-models (each fitted using occurrence and covariate data from within MBS) for each 5 km \times 5 km pixel within and outside MBS.

The model's predictive performance was evaluated using the area under the receiver operator curve (AUC) statistic, *i.e.* the area under a plot of the true positive rate versus false positive rate, reflecting the ability to discriminate between presence and background records [52]. The overall statistic was calculated as the mean of the AUCs for each of the 500 sub-models, calculated under 10-fold cross validation.

While each sub-model in the BRT ensemble was fitted using occurrence and background data from MBS (the model training region), the goal of our analysis was to predict to a much broader study area from northeast Bangladesh to southwest Papua New Guinea. To assess the model's predictive performance outside its training region, a separate AUC value was calculated for each sub-model using a validation dataset made up of presence and absence records from locations outside MBS. This AUC was calculated for each sub-model and then averaged across all 500 sub-models in the ensemble. Further information regarding the calculation of the AUC statistics is provided in the Supporting Information (<u>S1 File</u>).

Multivariate environmental similarity surface

The geographic regions outside MBS encompasses environments beyond the ranges of covariate values sampled within the training dataset. Model predictions to such environments are inherently less reliable than interpolations made to areas with environments within the range of covariate values in the training dataset. Thus it is important to assess the environmental similarities and differences between model training and prediction regions [53].

To investigate whether predictions to new geographic regions required extrapolation to covariate values beyond the range of the model training data, we computed and plotted a multivariate environmental similarity surface (MESS) [54] using R packages "dismo" [55] and "raster" [56]. This surface represents the similarity of the environment at each location to the covariate values at the presence and background locations in MBS (the reference data). The MESS calculation produces negative values for novel environments, locations where at least one covariate has a value that is outside the range of reference values (hereafter extrapolation), and positive values for locations within this range (hereafter interpolation). We converted the raw MESS output into a binary map indicating areas in which model predictions used interpolation versus extrapolation.

Mask

The model output was restricted to areas within the range of species known and hypothesized to be required for zoonotic transmission (*i.e.* the overlap in range maps of at least one reservoir and vector species), using previously reported species range maps [26].

A high resolution map for the zone of zoonotic transmission was also generated (<u>S3 Fig</u>) using existing species distribution maps and occurrence datasets [<u>26</u>]. Threshold

environmental suitability values for each of the species distribution maps for *M. fascicularis*, *M. nemestrina*, *M. leonina*, and the Leucosphyrus Group were determined to incorporate 90% of presence points in each species' respective occurrence database. We used these thresholds to classify each continuous species map as either present or absent for every $5 \text{ km} \times 5 \text{ km}$ pixel in the study region. These maps were combined to produce a final binary output showing areas of spatial co-occurrence of all species required for zoonotic, vector-borne transmission to humans *i.e.* presence of at least one macaque species plus at least one member of the Leucosphyrus Group.

Results

A total of 439 *P. knowlesi* occurrence records were identified, consisting of 301 presence and 138 absence records. The evaluation dataset (records falling outside MBS) totaled 131 records, comprising 29 point locations and 102 polygons (Fig 2A). The occurrence dataset used for model fitting (records falling within MBS) totaled 198 records, corresponding to 62 unique point locations and 136 polygons (Fig 2B). The model fitting dataset consisted of human (166), monkey (23) and mosquito (9) occurrence records.

The model predictions for the geographical variation in *P. knowlesi* infection risk in humans are displayed in Fig 3A. Overall, 10-fold cross validation statistics for the model ensemble (calculated using model training data) indicated high predictive performance with an AUC of 0.833 (SE \pm 0.002). A map of model uncertainty is displayed in the Supporting Information (S4 Fig).

The model output was restricted to areas within the geographic range of the species required for zoonotic transmission. An unmasked version of the mean output, showing relative suitability for zoonotic *P. knowlesi* transmission, is provided in the Supporting Information (S5 Fig).

The predicted map is presented alongside a projection of malaria eliminating countries in the year 2025 (Fig 3B) and together the two maps show countries where *P. knowlesi* transmission may persist after the human malarias are eliminated. The elimination projections, generated by the University of California San Francisco Global Health Group's Malaria Elimination Initiative, are based on current national and regional goals as well as recent epidemiological trends for the human malarias, principally *P. falciparum* and *P. vivax* [57].

Within MBS, our model predicted considerable spatial variation in risk of *P. knowlesi* infection, with areas ranging from relatively low risk to high risk predicted within Peninsular Malaysia, and both Sabah and Sarawak States of Malaysian Borneo (Fig 3A).

The model also predicts areas of high risk for *P. knowlesi* infection in a number of countries that are forecast to be malaria-free by 2025 (Malaysia, Cambodia, Thailand and Vietnam) as well as countries projected to be eliminating malaria (Myanmar, Laos, Indonesia and the Philippines) (Fig 3B). Large areas of high risk were predicted in Myanmar, Laos, Cambodia and Indonesia, with smaller areas predicted in Vietnam and Thailand. Human cases of *P. knowlesi* infection have been reported across this broad area (Fig 2A).

Regions for which we have no field data include areas of high predicted-risk, for example, eastern and western parts of Indonesia, and far eastern parts of India, although the predictions for the latter depend on whether *M. leonina*, included in the range of zoonotic transmission, is indeed a reservoir species ($\underline{Fig 3A}$).

Our predictions outside MBS are a result of both interpolation within the environmental space and extrapolation. The binary MESS map (Fig 3C) shows that model extrapolation to novel environments occurred in large regions in Cambodia, Vietnam, Thailand, Myanmar, India, and the Andaman and Nicobar Islands, indicating that predictions in these areas should be interpreted with caution. The model did, however, demonstrate high predictive performance at sites outside the model-training region, with an AUC statistic of 0.796 (SE \pm 0.003)

PLOS | NEGLECTED TROPICAL DISEASES



Fig 2. Occurrence data used for model fitting and evaluation. A. Location of presence and absence points/polygons outside Malaysia, Brunei and Singapore used for model evaluation. B. Location of presence and absence points/polygons as well as background points from Malaysia, Brunei and Singapore used for model fitting.

doi:10.1371/journal.pntd.0004915.g002

calculated using 131 presence/absence locations from the evaluation dataset ($\underline{Fig 2A}$). The predicted values for the evaluation data are presented in the Supporting Information ($\underline{S6 Fig}$).

The main predictors for *P. knowlesi* infection risk were urban accessibility, human population density, elevation, proportional cover of land with croplands and environmental suitability for *M. nemestrina*. Marginal effect plots for each of these covariates are displayed in the Supporting Information (S7 Fig).

Discussion

Using a niche model informed by a spatial database of *P. knowlesi* infections, a range of environmental and socio-economic data, and reservoir and vector species distributions, we have produced the first map of the predicted geographical distribution for *P. knowlesi* malaria. Empirical data on *P. knowlesi* presence or absence is lacking for most of Southeast Asia and this map provides an initial evidence base to prioritize areas for disease surveillance and future epidemiological investigations.

The predictive performance of the model was high and it also had a high capacity to predict suitability for infection in regions outside MBS. The latter result should, however, be treated with caution as data for model evaluation was only available from a limited number of locations outside MBS, and the selection of locations for which *P. knowlesi* has been tested is likely to be biased.

Another important caveat is the large area to which model predictions were made, relative to the model training region, since this required the model to extrapolate to some novel environments (see Fig 3C). Extrapolated predictions are inherently less reliable than those made in areas of interpolation and include large parts of continental Asia. Sampling for *P. knowlesi* infections in areas of extrapolation is likely to have the biggest impact on improving the disease risk predictions.

The final map therefore shows the risk of zoonotic *P. knowlesi* transmission from known reservoirs (specifically *M. nemestrina* and *M. fascicularis*) and vectors of the *Anopheles leuco-sphyrus* Group. If human-to-human transmission were occurring, this form of the disease is likely to have a different niche to the zoonotic disease, *i.e.* a different relationship with environmental, socioeconomic and biological factors. Thus our model is not appropriate to predict human-to-human transmission risk.

It is also important to note that the limits of zoonotic transmission, within which we have predicted infection risk, were defined using the reservoir and vector ranges generated by our earlier work and these ranges reflect the fact that species distributions are not fixed. Specifically, these ranges included introduced populations of the two macaque species, for example, pet *M*. *fascicularis* and *M*. *nemestrina* macaques are commonly found on Sulawesi where the environment is predicted to be suitable for the establishment of feral populations [26]. The predictions for infection risk that we present here therefore include locations on this island.

Human *P. knowlesi* infections have been identified beyond the ranges of both *M. nemestrina* and *M. fascicularis. Macaca leonina*, whose range extends farther north into Myanmar where these human cases were reported [58], has thus been implicated as a putative host species. We allowed predictions within the range of *M. leonina* but since this species is not found in MBS, it was not used as an explanatory covariate for model fitting. This may have impacted model predictions in the most northern parts of our study area where the environmental suitability for



Fig 3. Maps of *Plasmodium knowlesi* malaria risk, human malaria elimination status, and model extrapolation versus interpolation. A. Predicted risk of *P. knowlesi* malaria ranging from low to high risk. **B.** Countries projected to be malaria-free, eliminating malaria, or controlling malaria by 2025 (Map sourced from the University of California San Francisco Global Health Group's Malaria Elimination Initiative) **C.** Comparison of environments in Malaysia, Brunei and Singapore (the model training region) with those across the rest of Southeast Asia, using all covariates and the multivariate environmental similarity surface (MESS) methods. The map distinguishes between areas of model interpolation and areas where the model was required to extrapolate to novel environments.

doi:10.1371/journal.pntd.0004915.g003

M. nemestrina and *M. fascicularis* is low, but high for *M. leonina*. Again, sampling in these areas, particularly northern Myanmar, would improve the predictions.

Furthermore, two distinct *P. knowlesi* parasite populations, linked to *M. nemestrina* and *M. fascicularis* respectively, have been identified in human patients from Malaysia [59]. It is reasonable to assume that only the strain associated with *M. fascicularis* is circulating and infecting humans in areas of continental Asia, where *M. nemestrina* is absent, and it may have a distinct relationship with environmental and socioeconomic variables compared to the mixture of parasite infections in patients from Malaysia. The presence of Leucosphyrus Complex vectors in Malaysia and Dirus Complex vectors in continental Asia [26] further adds to the possibility of different relationships between disease risk and the environment in these two regions.

Comparing our predicted *P. knowlesi* risk map (Fig 3A) with the map of current sampling efforts (Fig 2), and the map of malaria eliminating countries (Fig 3B), allows us to identify relative surveillance priorities for *P. knowlesi*. These include a number of regions in Thailand (Phisanulok, Phetchuban, Chaiyaphum, Prachan Buri, and southern Nakhon Ratchasima) and Vietnam (Lam Dong, Phu Yen, Gia Lai, and Kon Tum). We also propose that further surveillance in previously sampled areas of Thailand, Vietnam and Cambodia is required to fully understand the distribution of *P. knowlesi* in countries close to eliminating the human malarias.

Among the countries next expected to eliminate the human malarias, our results highlight a need for surveillance in un-sampled, high-risk areas in Myanmar, Laos, and Sumatra and Kalimantan in Indonesia. Initial studies have reported cases in Aceh on Sumatra [13], and South and Central Kalimantan [60, 61] but no published reports are available from the other parts of these regions. Further surveillance is also needed in previously sampled areas, including Palawan in the Philippines.

Importantly, our map predicts the environmental suitability for infection, not the prevalence of infection or incidence of cases in these places. The higher numbers of reported cases in Malaysia is not proof that the disease risk is higher here because most locations outside Malaysia simply have not been surveyed and *P. knowlesi* could be misdiagnosed as one of the human malarias. Studies that have investigated numbers of cases or infections have sampled a wide array of communities including malaria patients [4], patients diagnosed as *P. malariae* by microscopy [62], and whole communities [63], meaning the disease prevalence indicators generated are not directly comparable. Until more locations are surveyed using a consistent measurement (ideally infections in a cross section of the community) and diagnostics that distinguish all human malarias, we cannot draw any firm conclusions about relative disease prevalence [64]. Studies of other diseases have, however, found a relationship between the environmental suitability for infection and infection prevalence or case incidence [17, 65] and this is a potential use of future iterations of this map. It will be important to update the predictions presented here when new data become available, and systems are available to generate updated predictive maps [66]. Importantly the map presented here provides key information about the locations where new surveys for P. knowlesi infections would be most informative.

As the volume and quality of geographical data on *P. knowlesi* infections increases across Southeast Asia, these maps will iteratively improve. For now, the work presented here provides the best evidence-base currently available for prioritizing *P. knowlesi* surveillance to better understand its spatial distribution and its wider contribution to malaria cases.

Supporting Information

S1 File. Methods supplement. (DOCX)

S2 File. Database of *Plasmodium knowlesi* infections recorded in humans, macaque reservoirs and vectors.

(XLSX)

S1 Fig. Covariates used in predicting the distribution of risk of *Plasmodium knowlesi* **malaria. A.** Displays human population density. **B-D.** Show the relative environmental suitability for vector (the Leucosphyrus Group) and reservoir species (*Macaca fascicularis* and *M. nemestrina*) of *P. knowlesi*, respectively. **E.** Shows an index of temperature suitability for *P. falciparum* transmission. **F** and **G.** Display values for tasselled cap wetness, which is measure of surface moisture (mean and standard deviation, respectively). **H.** Displays standard deviation values for tasselled cap brightness, which is a measure of moisture on bare surfaces. **I.** Gives the time required to travel from each geographic location to a large city via land or water-based transport networks. **J.** Displays elevation. For details of how each of these covariates layers was derived see <u>S1 File</u>.

(TIF)

S2 Fig. Land cover covariates used in predicting the distribution of risk of *Plasmodium knowlesi* **malaria. A-I.** Displays proportional cover for 2012 of lands with croplands, croplands natural vegetation mosaics, open shrublands, permanent wetlands, grasslands, intact forest, disturbed forest, woody savannas and savannas, respectively. For details of how each of these covariates layers was derived see <u>S1 File</u>. (TIF)

S3 Fig. Fine-scale map of presumed limits of zoonotic *Plasmodium knowlesi* **transmission.** Areas of spatial co-occurrence of known or putative reservoir (at least one of *M. fascicularis*, *M. nemestrina* or *M. leonina*) and vector species (members of the Leucosphyrus Group) are indicated, as well as areas where either reservoir or vector species are absent. (TIF)

S4 Fig. Map of model uncertainty. Standard deviation values for each pixel were calculated across the model ensemble. Areas from lower to higher standard deviation values are shown. (TIF)

S5 Fig. Unmasked mean model output. Suitability for zoonotic *Plasmodium knowlesi* transmission from known reservoir and vector species from relative low to high suitability. (TIF)

S6 Fig. Predicted disease risk values at locations with confirmed/unconfirmed presence and absence from outside Malaysia, Brunei and Singapore. The black dots represent the predicted values of confirmed/unconfirmed *P. knowlesi* presence and absence points and violin plots showing the density of points at each predicted value are shown in grey. Reports of *P. knowlesi* that were not supported by results from a second independent group working in the same region were classified as unconfirmed. (TIF)

S7 Fig. Marginal effect plots for the most influential covariates. The black line represents the mean marginal effect and grey envelopes the associated 95% quantiles. The mean relative contribution is displayed in the top left corner of each plot. (TIF)

Acknowledgments

We thank Maria Devine for proofreading the manuscript.

Author Contributions

Conceived and designed the experiments: FMS NG CLM. Performed the experiments: FMS. Analyzed the data: FMS NG CLM. Contributed reagents/materials/analysis tools: ZH DJW AW HSG KEB CP SJ YLL MM RA SB PWG IRFE IV BS CLM. Wrote the paper: FMS DMP OJB SIH BS NG CLM.

References

- 1. William T, Rahman HA, Jelip J, Ibrahim MY, Menon J, Grigg MJ, et al. Increasing incidence of *Plasmodium knowlesi* malaria following control of *P. falciparum* and *P. vivax* malaria in Sabah, Malaysia. PLoS Negl Trop Dis. 2013; 7(1).
- 2. World Health Organization. World Malaria Report 2014 Geneva: World Health Organization; 2014. Available from: <u>http://www.who.int/malaria/publications/world_malaria_report_2014/en/</u>.
- Barber BE, William T, Grigg MJ, Menon J, Auburn S, Marfurt J, et al. A prospective comparative study of knowlesi, falciparum, and vivax malaria in Sabah, Malaysia: High proportion with severe disease from *Plasmodium knowlesi* and *Plasmodium vivax* but no mortality with early referral and artesunate therapy. Clin Infect Dis. 2013; 56(3):383–97. doi: <u>10.1093/cid/cis902</u> PMID: <u>23087389</u>
- Cox-Singh J, Davis TM, Lee KS, Shamsul SS, Matusop A, Ratnam S, et al. *Plasmodium knowlesi* malaria in humans is widely distributed and potentially life threatening. Clin Infect Dis. 2008; 46(2):165– 71. doi: <u>10.1086/524888</u> PMID: <u>18171245</u>
- Nakaviroj S, Kobasa T, Teeranaipong P, Putaporntip C, Jongwutiwes S. An autochthonous case of severe *Plasmodium knowlesi* malaria in Thailand. Am J Trop Med Hyg. 2015; 92(3):569–72. doi: <u>10.</u> <u>4269/ajtmh.14-0610</u> PMID: <u>25535314</u>
- Seilmaier M, Hartmann W, Beissner M, Fenzl T, Haller C, Guggemos W, et al. Severe *Plasmodium* knowlesi infection with multi-organ failure imported to Germany from Thailand/Myanmar. Malar J. 2014; 13:422. doi: 10.1186/1475-2875-13-422 PMID: 25367021
- 7. Azidah AK, Faizal MMA, Lili HY, Zeehaida M. Severe *Plasmodium knowlesi* infection with multiorgan involvement in north east peninsular Malaysia. Trop Biomed. 2014; 31(1):31–5. PMID: <u>24862042</u>
- Singh B, Daneshvar C. Human infections and detection of *Plasmodium knowlesi*. Clin Microbiol Rev. 2013; 26(2):165–84. doi: <u>10.1128/CMR.00079-12</u> PMID: <u>23554413</u>
- Yusof R, Lau YL, Mahmud R, Fong MY, Jelip J, Ngian HU, et al. High proportion of knowlesi malaria in recent malaria cases in Malaysia. Malar J. 2014; 13:168. doi: <u>10.1186/1475-2875-13-168</u> PMID: <u>24886266</u>
- 10. Goh XT, Al Lim Y, Vythilingam I, Chew CH, Lee PC, Ngui R, et al. Increased detection of *Plasmodium knowlesi* in Sandakan division, Sabah as revealed by PlasmoNex (TM). Malar J. 2013; 12:28.
- Barber BE, William T, Grigg MJ, Yeo TW, Anstey NM. Limitations of microscopy to differentiate Plasmodium species in a region co-endemic for Plasmodium falciparum, Plasmodium vivax and Plasmodium knowlesi. Malar J. 2013; 12.
- Moyes CL, Henry AJ, Golding N, Huang Z, Singh B, Baird JK, et al. Defining the geographical range of the *Plasmodium knowlesi* reservoir. PLoS Negl Trop Dis. 2014; 8(3):e2780. doi: <u>10.1371/journal.pntd.</u> 0002780 PMID: 24676231
- Coutrier F, Cotter C, Tirta Y, Schwartz A, Zarlinda I, Basri H, et al. Serial molecular identification to confirm the presence of Plasmodium knowlesi in Indonesia. Poster session presented at: 64th Annual Meeting of The American Society for Tropical Medicine and Hygiene; 2015 Oct 25–29; Philadelphia, PA
- Ng OT, Ooi EE, Lee CC, Lee PJ, Ng LC, Pei SW, et al. Naturally acquired human *Plasmodium knowlesi* infection, Singapore. Emerg Infect Dis. 2008; 14(5):814. doi: <u>10.3201/eid1405.070863</u> PMID: <u>18439370</u>
- Cotter C, Sturrock HJW, Hsiang MS, Liu J, Phillips AA, Hwang J, et al. The changing epidemiology of malaria elimination: new strategies for new challenges. Lancet. 2013; 382(9895):900–11. doi: <u>10.1016/</u> <u>S0140-6736(13)60310-4</u> PMID: <u>23594387</u>
- Vythilingam I. Plasmodium knowlesi and Wuchereria bancrofti: their vectors and challenges for the future. Front Physiol. 2012; 3:115. doi: 10.3389/fphys.2012.00115 PMID: 22557977
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496(7446):504–7. doi: <u>10.1038/nature12060</u> PMID: <u>23563266</u>
- 18. Pigott DM, Bhatt S, Golding N, Duda KA, Battle KE, Brady OJ, et al. Global distribution maps of the leishmaniases. eLife. 2014; 3:e02851.

- Pigott DM, Golding N, Mylne A, Huang Z, Henry AJ, Weiss DJ, et al. Mapping the zoonotic niche of Ebola virus disease in Africa. eLife. 2014; 3:e04395. doi: <u>10.7554/eLife.04395</u> PMID: <u>25201877</u>
- Mylne AQ, Pigott DM, Longbottom J, Shearer F, Duda KA, Messina JP, et al. Mapping the zoonotic niche of Lassa fever in Africa. Trans R Soc Trop Med Hyg. 2015; 109(8):483–92. doi: <u>10.1093/trstmh/</u> <u>trv047</u> PMID: <u>26085474</u>
- Pigott DM, Golding N, Mylne A, Huang Z, Weiss DJ, Brady OJ, et al. Mapping the zoonotic niche of Marburg virus disease in Africa. Trans R Soc Trop Med Hyg. 2015.
- 22. Messina JP, Pigott DM, Golding N, Duda KA, Brownstein JS, Weiss DJ, et al. The global distribution of Crimean-Congo hemorrhagic fever. Trans R Soc Trop Med Hyg. 2015; 109(8):503–13. doi: <u>10.1093/</u> <u>trstmh/trv050</u> PMID: 26142451
- Messina JP, Kraemer MUG, Brady OJ, Pigott DM, Shearer FM, Weiss DJ, et al. Mapping global environmental suitability for Zika virus. eLife. 2016; 5:e15272. doi: 10.7554/eLife.15272 PMID: 27090089
- Elith J, Graham CH, Anderson RP, Dudik M, Ferrier S, Guisan A, et al. Novel methods improve prediction of species' distributions from occurrence data. Ecography. 2006; 29(2):129–51.
- Reisen WK. Landscape epidemiology of vector-borne diseases. Annu Rev Entomol. 2010; 55(1):461– 83.
- Moyes CL, Shearer FM, Huang Z, Wiebe A, Gibson HS, Nijman V, et al. Predicting the geographical distributions of the macaque hosts and mosquito vectors of *Plasmodium knowlesi* malaria in forested and non-forested areas. Parasit Vectors. 2016; 9(1):242. doi: <u>10.1186/s13071-016-1527-0</u> PMID: <u>27125995</u>
- Jongwutiwes S, Buppan P, Kosuvin R, Seethamchai S, Pattanawong U, Sirichaisinthop J, et al. *Plasmodium knowlesi* malaria in humans and macaques, Thailand. Emerg Infect Dis. 2011; 17(10):1799–806. doi: <u>10.3201/eid1710.110349</u> PMID: <u>22000348</u>
- Lee KS, Divis PC, Zakaria SK, Matusop A, Julin RA, Conway DJ, et al. *Plasmodium knowlesi*: reservoir hosts and tracking the emergence in humans and macaques. PLoS Pathog. 2011; 7(4):e1002015. doi: <u>10.1371/journal.ppat.1002015</u> PMID: <u>21490952</u>
- Divis PCS, Singh B, Anderios F, Hisam S, Matusop A, Kocken CH, et al. Admixture in humans of two divergent *Plasmodium knowlesi* populations associated with different macaque host species. PLoS Path. 2015; 11(5):e1004888.
- Assefa S, Lim C, Preston MD, Duffy CW, Nair MB, Adroub SA, et al. Population genomic structure and adaptation in the zoonotic malaria parasite *Plasmodium knowlesi*. Proc Natl Acad Sci USA. 2015; 112 (42):13027–32. doi: <u>10.1073/pnas.1509534112</u> PMID: <u>26438871</u>
- Faust C, Dobson AP. Primate malarias: diversity, distribution and insights for zoonotic *Plasmodium*. One Health. 2015; 1:66–75.
- Marchand RP, Culleton R, Maeno Y, Quang NT, Nakazawa S. Co-infections of *Plasmodium knowlesi*, *P. falciparum*, and *P. vivax* among humans and *Anopheles dirus* mosquitoes, Southern Vietnam. Emerg Infect Dis. 2011; 17(7):1232–9. doi: 10.3201/eid1707.101551 PMID: 21762577
- Jiram AI, Vythilingam I, NoorAzian YM, Yusof YM, Azahari AH, Fong M-Y. Entomologic investigation of *Plasmodium knowlesi* vectors in Kuala Lipis, Pahang, Malaysia. Malar J. 2012; 11:213. PMID: 22727041
- 34. Wong M, Chua T, Leong C, Khaw LT, Fornace K, Wan-Sulaiman W-Y, et al. Seasonal and spatial dynamics of the primary vector of *Plasmodium knowlesi* within a major transmission focus in Sabah, Malaysia. PLoS Negl Trop Dis. 2015; 9(10):e0004135. doi: <u>10.1371/journal.pntd.0004135</u> PMID: <u>26448052</u>
- Vythilingam I, Lim YAL, Venugopalan B, Ngui R, Leong CS, Wong ML, et al. *Plasmodium knowlesi* malaria an emerging public health problem in Hulu Selangor, Selangor, Malaysia (2009–2013): epidemiologic and entomologic analysis. Parasit Vectors. 2014; 7:436. doi: <u>10.1186/1756-3305-7-436</u> PMID: <u>25223878</u>
- Phillips S, Dudik M, Elith J, Graham C, Lehmann A, Leathwick J, et al. Sample selection bias and presence-only distribution models: implications for background and pseudo-absence data. Ecol Appl. 2009; 19:181–97. PMID: <u>19323182</u>
- World Health Organization. World Malaria Report 2012 Geneva: World Health Organization; 2012. Available from: <u>http://www.who.int/malaria/publications/world_malaria_report_2012/en/</u>.
- Friedl MA, Sulla-Menashe D, Tan B, Schneider A, Ramankutty N, Sibley A, et al. MODIS Collection 5 global land cover: Algorithm refinements and characterization of new datasets. Remote Sens Environ. 2010; 114(1):168–82.
- Potapov P, Yaroshenko A, Turubanova S, Dubinin M, Laestadius L, Thies C, et al. Mapping the world's intact forest landscapes by remote sensing. Ecol Soc. 2008; 13(2).

- 40. Farr TG, Rosen PA, Caro E, Crippen R, Duren R, Hensley S. The Shuttle Radar Topography Mission. Rev Geophys. 2007; 45:RG2004.
- Gething PW, Van Boeckel TP, Smith DL, Guerra CA, Patil AP, Snow RW, et al. Modelling the global constraints of temperature on transmission of *Plasmodium falciparum* and *P. vivax*. Parasit Vectors. 2011; 4(1):1–11.
- Lobser S, Cohen W. MODIS tasselled cap: land cover characteristics expressed through transformed MODIS data. Int J Remote Sens. 2007; 28:5079–101.
- Weiss DJ, Atkinson PM, Bhatt S, Mappin B, Hay SI, Gething PW. An effective approach for gap-filling continental scale remotely sensed time-series. ISPRS J Photogram Remote Sens. 2014; 98:106–18.
- 44. WorldPop. Gridded Population Distributions. 2015 [Accessed: July 2015]. Available from: <u>http://www.worldpop.org.uk/data/summary/?contselect=Asia&countselect=Whole+Continent&typeselect=</u> Population+2010.
- 45. Center for International Earth Science Information Network—CIESIN—Columbia University, United Nations Food + Agriculture Programme—FAO, Centro Internacional de Agricultura Tropical—CIAT. Gridded Population of the World, Version 4 (GPWv4): Population Count Grid. Palisades, NY: NASA Socioeconomic Data and Applications Center (SEDAC); 2005.
- Joint Research Center Global Environmental Modelling Unit. Travel time to major cities: A global map of accessibility. Available from: <u>http://bioval.jrc.ec.europa.eu/products/gam/sources.htm</u>.
- Ridgeway G with contributions from others (2015) gbm: generalized boosted regression models. R package version 2.1.1. Available from: <u>http://CRAN.R-project.org/package=gbm</u>.
- Elith J, Leathwick JR, Hastie T. A working guide to boosted regression trees. J Anim Ecol. 2008; 77 (4):802–13. doi: <u>10.1111/j.1365-2656.2008.01390.x</u> PMID: <u>18397250</u>
- Hansen MC, Potapov PV, Moore R, Hancher M, Turubanova SA, Tyukavina A, et al. High-resolution global maps of 21st-century forest cover change. Science. 2013; 342(6160):850–3. doi: <u>10.1126/</u> science.1244693 PMID: 24233722
- Fornace KM, Drakeley CJ, William T, Espino F, Cox J. Mapping infectious disease landscapes: unmanned aerial vehicles and epidemiology. Trends Parasitol. 2014; 30(11):514–9. doi: <u>10.1016/j.pt.</u> 2014.09.001 PMID: 25443854
- Barbet-Massin M, Jiguet F, Albert CH, Thuiller W. Selecting pseudo-absences for species distribution models: how, where and how many? Methods Ecol Evol. 2012; 3(2):327–38.
- 52. Fleiss J, Levin B, Paik M. Statistical methods for rates and proportions. Hoboken, New Jersey: John Wiley & Sons; 2003.
- Elith J, Leathwick JR. Species distribution models: ecological explanation and prediction across space and time. Annu Rev Ecol Evol Syst. 2009; 40(1):677–97.
- Elith J, Kearney M, Phillips S. The art of modelling range-shifting species. Methods Ecol Evol. 2010; 1 (4):330–42.
- Hijmans RJ, Phillip S, Leathwick J, Elith J (2016) dismo: species distribution modeling. R package version 1.0–15. Available from: <u>http://CRAN.R-project.org/package=dismo</u>.
- Hijmans RJ (2015) raster: geographic data analysis and modeling. R package version 2.5.2. Available from: <u>http://CRAN.R-project.org/package=raster</u>.
- Newby G, Bennett A, Larson E, Cotter C, Shretta R, Phillips AA, et al. The path to eradication: a progress report on the malaria-eliminating countries. Lancet. 2016; 387(10029):1775–84. doi: <u>10.1016/S0140-6736(16)00230-0</u> PMID: <u>27116283</u>
- 58. Zhu H, Li J, Zheng H. Human natural infection of *Plasmodium knowlesi*. Chin J Parasitol Parasit Dis. 2006; 24(1):70.
- Divis PCS, Singh B, Anderios F, Hisam S, Matusop A, Kocken CH, et al. Admixture in Humans of Two Divergent Plasmodium knowlesi Populations Associated with Different Macaque Host Species. PLoS Path. 2015; 11(5).
- Sulistyaningsih E, Fitri LE, Löscher T, Berens-Riha N. Diagnostic difficulties with *Plasmodium knowlesi* infection in humans. Emerg Infect Dis. 2010; 16(6):1033. doi: <u>10.3201/eid1606.100022</u> PMID: <u>20507769</u>
- **61.** Hadi MS, Sumedi DP. Malaria Baru Dari Primata [New Primate Malaria]. Jakarta: Pusat Komunikasi Publik Setjen Kementerian Kesehatan Ri, 2013.
- Vythilingam I, Noorazian YM, Huat TC, Jiram AI, Yusri YM, Azahari AH, et al. *Plasmodium knowlesi* in humans, macaques and mosquitoes in peninsular Malaysia. Parasit Vectors. 2008; 1(1):26. doi: <u>10.</u> <u>1186/1756-3305-1-26</u> PMID: <u>18710577</u>
- 63. Steenkeste N, Rogers WO, Okell L, Jeanne I, Incardona S, Duval L, et al. Sub-microscopic malaria cases and mixed malaria infection in a remote area of high malaria endemicity in Rattanakiri province,

Cambodia: implication for malaria elimination. Malar J. 2010; 9:108. doi: <u>10.1186/1475-2875-9-108</u> PMID: <u>20409349</u>

- 64. Fornace KM, Nuin NA, Betson M, Grigg MJ, William T, Anstey NM, et al. Asymptomatic and submicroscopic carriage of *Plasmodium knowlesi* malaria in household and community members of clinical cases in Sabah, Malaysia. J Infect Dis. 2016; 213(5):784–7. doi: <u>10.1093/infdis/jiv475</u> PMID: <u>26433222</u>
- Limmathurotsakul D, Golding N, Dance DAB, Messina JP, Pigott DM, Moyes CL, et al. Predicted global distribution of *Burkholderia pseudomallei* and burden of melioidosis. Nat Microbiol. 2016; 1:15008. PMID: <u>26877885</u>
- Patching HMM, Hudson LM, Cooke W, Garcia AJ, Hay SI, Roberts M, et al. A supervised learning process to validate online disease reports for use in predictive models. Big Data. 2015; 3(4):230–7. PMID: 26858916